Author's response to reviews

Title: Deletion of the TNFAIP3/A20 gene detected by FICTION analysis in classical Hodgkin lymphoma

Authors:

Junko Nomoto (jnomoto@ncc.go.jp)
Nobuhiro Hiramoto (nccnsct@gmail.com)
Motohiro Kato (katom-tsky@umin.ac.jp)
Masashi Sanada (sanada-tky@umin.net)
Akiko M Maeshima (akmaeshi@ncc.go.jp)
Hirokazu Taniguchi (hitani@ncc.go.jp)
Fumie Hosoda (fhosoda@ncc.go.jp)
Yoshitaka Asakura (yasakura@ncc.go.jp)
Wataru Munakata (wmunakat@ncc.go.jp)
Dai Maruyama (dmaruyama@ncc.go.jp)
Takashi Watanabe (takawata@ncc.go.jp)
Hitoshi Nakagama (hnakagama@ncc.go.jp)
Kengo Taniguchi (kentakeuchi-tky@umin.net)
Kensei Tobinai (ktobinai@ncc.go.jp)
Seishi Ogawa (sogawa-tky@umin.ac.jp)
Yukio Kobayashi (ykkobaya@ncc.go.jp)

Version: 2 Date: 7 June 2012

Author's response to reviews:

This is an update according to your suggestions. We had missed the sentences regarding using human materials. A new sentence was added as follows: 'This research was carried out in compliance with the Helsinki Declaration, and was approved by Institutional Review Board at NCC (20-010).'

And I noticed one of the co-author's name was mistaken. I am changing it, too.